Author's response to reviews

Title: Patterns of comorbidity in community-dwelling older people hospitalised for fall-related injury: A cluster analysis

Authors:

Trang Vu (trang.vu@monash.edu)
Caroline F Finch (caroline.finch@monash.edu)
Lesley Day (lesley.day@monash.edu)

Version: 3 Date: 18 August 2011

Author's response to reviews:

18 August 2011
Dear BMC Series Editorial Production Team

Formatting changes to the manuscript have been made as requested in your email dated 18th August 2011. Specifically, the changes are as follows.

1. Major revisions excepted by the Executive Editor have been incorporated into manuscript.
2. Shading and colours within tables have been removed.
3. Minor revisions excepted by the Executive Editor have also been incorporated into the manuscript.
4. 'Post-it' note-style comments have been removed from the manuscript.
5. "Introduction" has been changed to "Background".
6. All highlighting/tracking has been removed from the manuscript.
7. Vertical lines have been removed from tables.
8. The footer "Trang Vu Page 1 8/08/2011" has been removed from the figures.
9. The footer has been removed from the manuscript.

I can be contacted at trang.vu@monsh.edu should you require further information.

Yours sincerely
Trang Vu
Corresponding author